Author’s response to reviews

Title: Trends in Immunization Completion and Disparities in the Context of Health Reforms: The case study of Tanzania.

Authors:

   Innocent A Semali (isemali@muhas.ac.tz)

Version: 6 Date: 16 October 2010

Author’s response to reviews: see over
To

Chief Editor,
BMC: Health System Research Journal,
BMC Journals

Dear Sirs,

I am kindly re-submitting my paper to you to kindly consider it for publication in the Health System Research Journal. The title has been edited and changed to: **Trends in Immunization Completion and Disparities in the Context of Health Reforms: The case study of Tanzania.**

The abstract had been edited please see track changes in the attached version.

All sections of the manuscript has been edited please see the tack changes of the attached version.

The corresponding author’s particulars remain as follows:
Name: Innocent A Semali
Address: School of Public Health and Social Sciences
Muhimbili University of Health and Allied Sciences
P.O. Box 65015
Dar es Salaam
Tanzania.
Telephone: +255(0)754269838; email: isemali@muhas.ac.tz.innosemali@yahoo.com

With kind regards.

I A Semali